

EO-1 Weekly Report


	Earth Observing – 1

	February 8, 2012 – February 14, 2012

	DOY
	039
	–
	045

	Mission Day
	4157
	–
	4163


EO-1 SPACECRAFT

Instruments

· Scheduled 104 science Data Collection Events (DCEs) this week 

· Received all images
· Performed Lunar Calibration on 039/08:19z

· Performed instrument decontamination cycles

· Conducted HSI deicing from 044/00:25z to 044/15:15z

· Conducted ALI outgassing from 044/00:35z to 044/15:25z
· Performed Instrument calibration

· Solar Array Characterization at 044/10:26z

· HSI solar calibration 044/21:57z
EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintained nominal spacecraft state of health
· The Lunar Calibration was performed

· This Lunar Calibration consisted of the 4 ALI/Hyperion scans but did not contain the final Atmospheric Corrector scan. Instead, after the 4th scan EO-1 will go back to Earth acquisition to reduce power usage.

· Despite the power reduction of removal of the 5th scan, the EO-1 Battery still reached 25.6 V, only .1 V over the TSM that would put the spacecraft into safe-hold 

· Due to this low level of power we will be postponing the Lunar Calibrations until we can determine a way to perform them without reaching that limit

· Also, after the lunar calibration the battery has not been charging to its 99.5+% at end of day

· There are some passes where it reaches this mark, but the majority of passes it only reaches 98.7-98.9%

· Talked with the Power people and suggested returning the VT level to 4.5 since we have seen the differential voltage level off. They agreed that it would be fine to return EO-1 to VT 4.5 and then we could keep it there or lower it again once we are back to the proper EOD SOC.
· The battery differential voltage appeared stay within the level range we have been seeing of -180mV to -210mV

· Monitoring the battery differential voltage, temperature, current, voltage and state of charge

· Since the battery Differential Voltage has appeared to level off in the -180mV to -210mV range, we believe that if we raise the VT level back to 4.5 to fix the EOD SOC problem above that the differential voltage will remain in an acceptable range
· After the VT level change back to 4.5 we will monitor continue to monitor the differential voltage and determine whether we will keep the VT at 4.5 or return it to 4.0

· Working with Flight Software people for EO-1 to modify the TSM limits

· Flight Software people are creating a load to change the TSM limits since we have seemed to plateau on the differential voltage
· The TRMM spacecraft has the same type of battery, 50 AH Super NiCad, and routinely sees differential voltages of +/-500mV
· Cataloguing all of the systems / spare parts for EO-1 to help the property manager and prepare for the contract switch

· TrollSat (TR2) is now certified for S-Band passes and will be putting 2-3 per week in the schedule
· In the coming weeks we will be setting up X-band tests with TR2

· Continuing to monitor SGS passes to ensure they remain stable
· SGS has an equipment problem and will be down until approximately August 2012
· SGS passes being taken on the SG-1/SG-22 hybrid antenna that was used during last equipment malfunction

Mission Planning
· Creating TDRS supports to monitor battery differential
· Looking at the steps required in creating a South Pole image

· Remaking the SOP for mission planning activities for ASPEN and ASIST SCP
Flight Dynamics
· Changed the TDRS names to ensure correct TDRS usage

· Monitored the GPS vs Ephemeris to make sure the change in script did what we expected and corrected the cross track errors that occurred with the GPS leap second broadcast 
· Worked with FDF for getting products from their new Product Servers

· Finalizing certificate based SFTP with new product server

· Looking at the steps required to take a South Pole image
System Administration
· Completed correction of DNS settings on the primary Flight Dynamics computer. This computer now points to another IONet DNS system as a backup instead of the CNE.

· Creating scripts to simplify the process for viewing firewall logs. These scripts will use the tags established in the firewall ruleset to narrow down the logs for quicker analysis.

· Increased free space on the backup ASIST T&C system’s root partition by deleting unnecessary files/folders. This partition is now below the disk space usage script’s warning threshold.

· Completed DAR related data collection request. The computer, user, and operating system information was entered in the spreadsheet and sent to the IT security group

· Completed creation of a thorough property spreadsheet. The systems in the Flight Software Lab, library, and offices were added to the list.

· Completed compiling a list of necessary spare parts and materials for the upcoming contract year. This list was sent to the FOT Lead for review and possible additions.

· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, and MOC CNE computer.

GROUND AND SPACE NETWORK

Station Downtimes 

· SGS has had an equipment malfunction and latest indication is that it will be back up next August
· SG1/22 has a communication line problem, workaround is in place, fix timing unknown
· HGS is down, unknown time to be back up

Operational Discrepancies

· PF1 pass on 040/23:06z had 5 scenes that at first appeared to be lost due to incomplete files and hundreds of sequence breaks in the X-band data.  After a playback from the ground station we received files that could be processed and were able to recover all 5 scenes
· 5 scenes recovered
Number of Real-Time Supports
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8-Feb-12 39

5 8 0 0

13

9-Feb-12 40

5 3 1 0

9

10-Feb-12 41

6 3 1 0

10

11-Feb-12 42

6 5 1 1

13

12-Feb-12 43

5 4 1 0

10

13-Feb-12 44

1 8 0 0

9

14-Feb-12 45

4 4 1 0

9

32 35 5 1 73 Weekly Totals


UPCOMING EVENTS

· VT level change back to 4.5

· TSM update onboard spacecraft
· TR2 X-Band pass testing
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		8-Feb-12		39		5		8		0		0		13

		9-Feb-12		40		5		3		1		0		9

		10-Feb-12		41		6		3		1		0		10

		11-Feb-12		42		6		5		1		1		13

		12-Feb-12		43		5		4		1		0		10

		13-Feb-12		44		1		8		0		0		9

		14-Feb-12		45		4		4		1		0		9

		Weekly Totals				32		35		5		1		73





Sheet2

		





Sheet3

		






